CS486: Spring 1999

Problem Set 10 Optional

Reading in Suppes:

Recall Chapters 1 – 3 and pages 159 – 181. Think of these as a study guide for set theory. Concentrate on the material from Suppes that was also covered in lecture. Only that material will be on the final exam. Also relate the Suppes material to what you learned about sets in CS280.

Suggested Optional Exercises

• p. 30, ex. 10 • p. 33, ex. 2, 4 • p. 46, ex. 14 • p. 56, ex. 4 (read p. 56 carefully) • p. 67, ex. 3, 6 • p. 86, ex. 1, 5 • p. 181, ex. 1, 3 • p. 189, ex. 10